import TypeList from "@site/src/components/TypeList"

# inventory

## Interfaces

- [FindConfig](../inventory/interfaces/inventory.FindConfig.mdx)
- [StringComparisonOperator](../inventory/interfaces/inventory.StringComparisonOperator.mdx)
- [NumericalComparisonOperator](../inventory/interfaces/inventory.NumericalComparisonOperator.mdx)
- [FilterableReservationItemProps](../inventory/interfaces/inventory.FilterableReservationItemProps.mdx)
- [FilterableInventoryItemProps](../inventory/interfaces/inventory.FilterableInventoryItemProps.mdx)
- [CreateInventoryItemInput](../inventory/interfaces/inventory.CreateInventoryItemInput.mdx)
- [CreateReservationItemInput](../inventory/interfaces/inventory.CreateReservationItemInput.mdx)
- [FilterableInventoryLevelProps](../inventory/interfaces/inventory.FilterableInventoryLevelProps.mdx)
- [CreateInventoryLevelInput](../inventory/interfaces/inventory.CreateInventoryLevelInput.mdx)
- [UpdateInventoryLevelInput](../inventory/interfaces/inventory.UpdateInventoryLevelInput.mdx)
- [BulkUpdateInventoryLevelInput](../inventory/interfaces/inventory.BulkUpdateInventoryLevelInput.mdx)
- [UpdateReservationItemInput](../inventory/interfaces/inventory.UpdateReservationItemInput.mdx)
- [IInventoryService](../inventory/interfaces/inventory.IInventoryService.mdx)
- [JoinerServiceConfigAlias](../inventory/interfaces/inventory.JoinerServiceConfigAlias.mdx)
- [JoinerServiceConfig](../inventory/interfaces/inventory.JoinerServiceConfig.mdx)
- [IModuleService](../inventory/interfaces/inventory.IModuleService.mdx)
- [SharedContext](../inventory/interfaces/inventory.SharedContext.mdx)

## Type Aliases

- [InventoryItemDTO](../inventory/types/inventory.InventoryItemDTO.mdx)
- [ReservationItemDTO](../inventory/types/inventory.ReservationItemDTO.mdx)
- [InventoryLevelDTO](../inventory/types/inventory.InventoryLevelDTO.mdx)
- [JoinerRelationship](../inventory/types/inventory.JoinerRelationship.mdx)
- [ModuleJoinerConfig](../inventory/types/inventory.ModuleJoinerConfig.mdx)
- [ModuleJoinerRelationship](../inventory/types/inventory.ModuleJoinerRelationship.mdx)
